Author’s response to reviews

Title: Oncological safety of intrafascial nerve-sparing radical prostatectomy compared with conventional process: a pooled review and meta-regression analysis based on available studies

Authors:
Xiao Wang (feiyao1943@163.com)
Yiqi Wu (reachiwu@qq.com)
Jia Guo (1073215852@qq.com)
Hui Chen (43671716@qq.com)
Xiaodong Weng (12942169@qq.com)
Xiuheing Liu (prof_liuxhmd@163.com)

Version: 2 Date: 01 May 2019

Author’s response to reviews:

Sorry for the delay of revision, due to that we mistaken the term for May 5th! We have revised the paper according to the editor's comments and a new version has been uploaded. If any other problems exist, please contact us.